Author's response to reviews

Title: Associations between Serotonin Transporter Gene Polymorphisms and Heat Pain Perception in Adults with Chronic Pain

Authors:

W. Michael Hooten (hooten.william@mayo.edu)
William R Hartman (hartman.william@mayo.edu)
John L Black (black.john@mayo.edu)
Heidi J Laures (laures.heidi@mayo.edu)
Denise L Walker (walker.denise@mayo.edu)

Version: 3 Date: 12 July 2013

Author's response to reviews: see over
July 12, 2013

Re: Revisions to manuscript # 3563893468040814

Professor Thomas Klopstock
BMC Medical Genetics

Dear Prof. Klopstock:

We would like to thank the editorial staff for the thoughtful review of our manuscript. As requested, the below information has been included after the discussion section of the manuscript. In the manuscript, the newly incorporated information has been highlighted in red for easy identification.

Competing interests
The authors state that they have no financial or non-financial competing interests.

Authors’ contributions
WMH, WRH and JLB designed the study. WMH, WRH, JLB, HJL and DLW performed the data analysis. WMH, WRH, JLB, HJL and DLW wrote the manuscript. All authors critically revised the manuscript and approved the final form of the manuscript. All authors read and approved the final manuscript.

Acknowledgements
The authors would like to thank the staff members of the Mayo Pain Rehabilitation Center and the Translational Research Unit for Chronic and Acute Pain.

Again, we would like to thank the editorial staff for their review of our work.

Sincerely,

W. Michael Hooten, MD